Author’s response to reviews

Title: Are variations in rates of attending cultural activities associated with population health in the United States?

Authors:

Anna V Wilkinson (awilkins@mdanderson.org)
Andrew J Waters (ajwaters@mdanderson.org)
Lars O Bygren (lars-olov.bygren@prevnut.ki.se)
Alvin R Tarlov (atarlov@gmail.com)

Version: 8 Date: 17 July 2007

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team:

Re: Submission 7899059451173832

I have highlighted the changes I made to the title page below:

Anna V Wilkinson¹§, Andrew J Waters², Lars Olov Bygren³⁴, Alvin R Tarlov⁵⁶

¹ Department of Epidemiology, Unit 1340, The University of Texas MD Anderson Cancer Center, 1155 Hermann Pressler Boulevard, Houston, TX 77030, United States
² Department of Behavioral Science, Unit 1330, The University of Texas MD Anderson Cancer Center, 1155 Hermann Pressler Boulevard, Houston, TX 77030, United States
³ Department of Biosciences, Karolinska Institute, SE-171 77, Stockholm, Sweden
⁴ Department of Community Health and Rehabilitation, University of Umeå, SE-901 87, Umeå, Sweden
⁵ James A. Baker III Institute for Public Policy, Rice University, 6100 Main Street, Houston, TX 77005, United States
⁶ Current affiliation: Department of Medicine, University of Chicago, 540 North State Street, #3801, Chicago, Il 60610, United States

In addition I changed “ABSTRACT” to “Abstract.”

I checked the rest of the formatting, and assuming I understand BMC’s formatting requirements, the remainder of the manuscript looks to me like it meets your standards. I hope it does. Please advice if not and I will make the necessary adjustments.

Again, thank you for your help and patience while working with us on this manuscript.

Sincerely,

Anna Wilkinson